Author's response to reviews

Title: Successful Removal of A Telephone Cable, A Self-Inserted Foreign Body through Urethra Into The Bladder: A case report

Authors:

Ravi K Trehan SpR (trehanravi@hotmail.com)
Athar Haroon SpR (atharharoon@yahoo.com)
Shaukat Memon (shaukat.memon@ulh.nhs.uk)
Derek Turner (derek.turner@ulh.nhs.uk)

Version: 4 Date: 30 August 2007

Author's response to reviews: see over
Date: 30th August 2007

Editorial Board
JMCR

Dear Sir / Ma’am

**Subject: MS: 1858107523150770**

Thank you for sending me comments of two reviewers and editorial board’s decision.

As advised by editorial board and peer reviewer 2, I have included following in revised manuscript,

1: A more up-to-date literature review of foreign bodies within the lower urinary tract.

2: Detailed description of various diagnostic and removal techniques, as recommended by peer reviewer 2.

3: I have also included further information about the case as advised by peer reviewer 1 that includes
   1. Patient received single adult dose of I/V Gentamicin.
   2. No sedation was used during the procedure
   3. Patient passed urine normally in A&E and then discharged.
   4. Patient failed to attend follow up cystoscopy appointment.
   5. Psychiatric referral was advised.

Please accept the revised manuscript.

If you need any further information please do not hesitate to contact me.

Thanking you

Sincerely yours

Mr Ravi Trehan
22a, York Road
Cheam
Surrey
SM2 6HH
trehanravi@hotmail.com